

# **Geophysical Research Letters**

## **RESEARCH LETTER**

10.1029/2018GL081363

#### **Key Points:**

- Equatorial Pacific sea surface temperature bias contributes to double ITCZ bias in climate models
- Interpretation of future tropical rainfall projections requires care as double ITCZ bias affects precipitation trends
- Improvement of bias in equatorial Pacific sea surface temperature may reduce uncertainty in rainfall projections

**Supporting Information:** 

Supporting Information S1

#### **Correspondence to:** D. Samanta,

dhruba@ntu.edu.sg

#### Citation:

Samanta, D., Karnauskas, K. B., & Goodkin, N. F. (2019). Tropical Pacific SST and ITCZ biases in climate models: Double trouble for future rainfall projections?. *Geophysical Research Letters*, 46, 2242–2252. https://doi.org/ 10.1029/2018GL081363

Received 17 NOV 2018 Accepted 30 JAN 2019 Accepted article online 1 FEB 2019 Published online 19 FEB 2019

©2019. The Authors.

This is an open access article under the terms of the Creative Commons Attribution-NoDerivs License, which permits use and distribution in any medium, provided the original work is properly cited and no modifications or adaptations are made. Dhrubajyoti Samanta<sup>1</sup>, Kristopher B. Karnauskas<sup>2,3</sup>, and Nathalie F. Goodkin<sup>1,4,5</sup>

<sup>1</sup>Asian School of the Environment, Nanyang Technological University, Singapore, <sup>2</sup>Cooperative Institute for Research in Environmental Sciences, University of Colorado Boulder, Boulder, Colorado, USA, <sup>3</sup>Department of Atmospheric and Oceanic Sciences, University of Colorado Boulder, Boulder, Colorado, USA, <sup>4</sup>Earth Observatory of Singapore, Singapore, <sup>5</sup>Department of Earth and Planetary Sciences, American Museum of Natural History, New York, New York, USA

**Abstract** The double Intertropical Convergence Zone bias remains a persistent problem in coupled general circulation model simulations. Due to the strong sea surface temperature (SST)-convection relationship in the tropics, precipitation biases are sensitive to background SST. Using historical simulations of 24 coupled general circulation models and an atmospheric general circulation model, we show that cold equatorial SST biases at least exacerbate double Intertropical Convergence Zone biases in the Pacific. A linear regression model is used to demonstrate that improved predictability of precipitation trends is possible with such model-dependent information as mean-state SST biases accompanying projected SST trends. These results provide a better understanding of the root of the double Intertropical Convergence Zone bias and a possible path to reduced uncertainty in future tropical precipitation trends.

**Plain Language Summary** Dozens of complex computer models of the climate system (accounting for both atmosphere and ocean) are used for climate change predictions over the coming decades as anthropogenic emissions of greenhouse gases persist. These computer models are essential yet imperfect. Two well-known mismatches (i.e., biases) in models that have persisted for many years include (1) too cold sea surface temperature (SST) along the equatorial Pacific Ocean and (2) excessive precipitation south of the equator, which appears as a double-peaked precipitation pattern known as the double Intertropical Convergence Zone (ITCZ). These are commonly referred to as the "cold tongue bias" and "double ITCZ bias," respectively. Our analysis confirms that they are closely related in models; the worse the cold tongue bias is, the more the ITCZ is split into two. We also found that predicted trends in tropical rainfall depend on these biases; models with the least severe SST bias will improve relative to today's climate as SST warms, whereas the double ITCZ in models with the most severe biases will remain the same even as equatorial SSTs warm. Finally, we demonstrate that rainfall predictions can be improved if information about biases is accounted for.

## **1. Introduction**

Convection and precipitation are strongly coupled over the tropics (Neelin & Held, 1987), such that uncertainties in resolving convection in coupled general circulation models (CGCMs) continue to result in a biased simulation of precipitation. These uncertainties include transitions between shallow and deep convection (Wang et al., 2015) and a smaller sea surface temperature (SST) threshold for triggering deep convection (Bellucci et al., 2010). Future changes of tropical Pacific precipitation are largely dependent on the response to changes in underlying SST (Power et al., 2013; Xie et al., 2010). Moreover, future precipitation changes in the tropical Pacific are controlled by two key processes—(i) a dynamic response from the overall slowing of tropical circulation and (ii) a "wet get wetter" thermodynamic response (Chadwick et al., 2013; Held & Soden, 2006; Seager et al., 2010; Vecchi & Soden, 2007). The underlying SST field can affect all these processes; therefore, the details of SST anomaly can largely impact precipitation changes (Chadwick et al., 2013; Kent et al., 2015).

Intertropical Convergence Zone (ITCZ) is the zonally oriented tropical precipitation band with the most intense rainfall globally (Schneider et al., 2014; Waliser & Gautier, 1993). Despite the seasonal cycle of solar radiation, the persistent displacement of the ITCZ north of the equator is a key asymmetric feature of global climate. Due to this hemispheric asymmetry, the annual-mean ITCZ lies ~5° north of the equator at most



**Figure 1.** Annual climatology, and biases in precipitation (in  $mm \cdot day^{-1}$ ), and sea surface temperature (SST; in °C) in historical simulations. Observed climatology of (a) precipitation and (b) SST; multimodel ensemble annual bias of (c) precipitation and (d) SST; and (e) linear regression of precipitation bias onto annual SST bias (in  $mm \cdot day^{-1} \cdot °C^{-1}$ ) averaged over 5°S–5°N, 160°E–120°W (EB) for 24 CMIP5 models. The sign of regression coefficient is reversed in (e). Black boxes in (c)–(e) denote the domain for the SST and precipitation bias calculations. Blue boxes in (c) and (e) denote northern and southern boxes for calculation of double-ITCZ bias.

longitudes, especially in the Pacific (Figure 1a). The annual mean northward displacement of the ITCZ is generally attributed to the excess annual mean heating of the northern hemisphere by the Atlantic meridional overturning circulation (Marshall et al., 2014), but is sensitive to regional SST anomalies (Adam, Schneider, et al., 2016).

The double ITCZ is a characterized as the proclivity of the CGCMs to produce excessive precipitation south of the equator, which, in the annual mean, gives the appearance of a nearly symmetric bimodal precipitation distribution straddling the equator (Adam et al., 2018; Lin, 2007; Zhang, 2001; Zhang et al., 2015), rather than a dominant peak north of the equator in observations. A southern ITCZ mostly occurs during boreal spring in the eastern Pacific, whereas, in the western Pacific, an excessive double ITCZ that straddles the equator is seen year-round (e.g., Adam et al., 2018). However, a northern ITCZ is observed in all seasons (Ma et al., 1996). A double ITCZ can also emerge in some climate models in the Atlantic sector, and generally in aqua planet simulations (e.g., Zhou & Xie, 2015). CGCMs in Coupled Model Intercomparison Project phase 5 (CMIP5) continue to suffer from the double ITCZ bias (e.g., Zhang et al., 2015), which has been linked to a multitude of factors, such as tropical circulation, future precipitation projections under climate change (Li & Xie, 2014), and climate sensitivity (Tian, 2015). Additionally, CGCMs simulate a cold tongue SST bias over the tropical Pacific (De Szoeke & Xie, 2008; Li & Xie, 2014; Richter et al., 2016), which is another persistent problem for a realistic simulation. However, due to the paucity of observations with sufficient temporal and spatial coverage, CGCMs remain a key tool for better understanding historical and future climates (Yang et al., 2018). The biases in cold tongue SST and ITCZ and the unclear relationship within them impose a major barrier to the realistic simulation of tropical precipitation trends in CGCMs not only over the tropical Pacific but also over the entire globe through, for example, Rossby wave responses to tropical heating (Gill, 1980; Trenberth et al., 1998). Moreover, the double ITCZ impacts the simulation and predictability of different modes of seasonal-to-interannual climate variability, such as the Madden-Julian Oscillation and the El Niño-Southern Oscillation (Guilyardi et al., 2003; Inness & Slingo, 2003; Wittenberg et al., 2006), limiting the certainty of CGCMs predictions, particularly the future trends in the face of anthropogenic radiative forcing.

The double-ITCZ bias, that is, excessive tropical precipitation south of the equator along with underestimated equatorial precipitation, leads to a major barrier for representing tropical climate dynamics in CGCM simulations (Bellucci et al., 2010; Lin, 2007; Oueslati & Bellon, 2015; Song & Zhang, 2018; Xiang et al., 2017). Several underrepresented dynamical and thermodynamical feedback, such as Bjerknes feedback (Li & Xie, 2014; Lin, 2007), SST-shortwave flux feedback (Lin, 2007), SST-latent heat flux feedback (Lin, 2007; Song & Zhang, 2016), and zonal SST gradients in the Pacific and Atlantic (Lin, 2007; Siongco et al., 2015), are known to affect the double-ITCZ bias in CGCM simulations. A double-ITCZ bias in CGCMs can emerge from several factors, such as the choice of convective parameterization scheme (Hirota et al., 2011), convective mixing (Möbis & Stevens, 2012; Song & Zhang, 2018), or from atmospheric transient eddies (Xiang et al., 2018). Additionally, a precipitation bias in CGCM simulations can emerge due to biases in tropical SST pattern associated with ocean current in the model itself (Wang et al., 2015). Indeed, Hwang and Frierson (2013) have shown that biases in cross-equatorial energy transport are linked to precipitation biases south of the equator. However, their suggestion that Southern Ocean low cloud biases are the cause of the biases in the cross-equatorial energy fluxes has since been shown to be unsupported (Adam et al., 2018; Hawcroft et al., 2017; Kay et al., 2016).

Because of structural diversity and different mean states of each CGCM, quantifying the origin of the double ITCZ remains a challenge. This also leads to a causal dilemma: is the double ITCZ a consequence of evolving SST biases or is the SST bias itself driven by double-ITCZ bias? While the later direction of causality may be important for some contexts, such as the study of cool climates with scarce precipitation (e.g., the Galapagos islands; Lietzke et al., 2001; Gu et al., 2005), we are driven here to investigate the possible role of the SST bias on the double-ITCZ bias. Earlier studies (e.g., Song & Zhang, 2016, 2018) investigated the southeastern Pacific SST bias on the double-ITCZ bias in one CGCM by prescribing local SST forcing. Few other studies (e.g., Ma et al., 1996; Qin & Lin, 2018) argued for an influence of marine low clouds. However, these studies neither investigated the direct response of SST forcing over the cold tongue region, nor studied SST response in a diverse ensemble of CGCMs.

In this study, we aim to investigate the relationship among mean-state biases in recent generation CGCMs, particularly between the double-ITCZ bias and the cold equatorial Pacific SST bias, as well as the implications for simulated tropical precipitation trends. Our study focuses on the aspects of year-round double-ITCZ bias present over central and western Pacific, however, does not involve seasonal double-ITCZ pattern over eastern Pacific or Atlantic sector during boreal spring.

### 2. Models and Methods

#### 2.1. Observations and CMIP5 Models

Monthly SST and precipitation output from historical runs for the period 1861–2005 from 24 CMIP5 models (Taylor et al., 2012) are employed in this study. A detailed list of CMIP5 models is provided in Table S1. For annual mean bias calculations, we used monthly precipitation data from the Global Precipitation Climatology Project (Adler et al., 2003) and SST from COBE SST version 2 (https://www.esrl.noaa.gov/psd/data/gridded/data.cobe2.htm). Annually resolved data for these variables were then transformed into a common grid of  $2.0^{\circ} \times 1.5^{\circ}$  using bilinear interpolation for calculating multimodel composites. Biases and trends in SST and precipitation are calculated over the equatorial Pacific box (EB; 5°S–5°N, 160°E–120°W).

The inadequacy of clear definitions and the use of various indicators in different studies to characterize the double-ITCZ bias have made understanding the key issues to address problematic (Popp & Lutsko, 2017). Furthermore, the earlier defined hemispheric asymmetry index, equatorial precipitation index (e.g., Adam, Bischoff, et al., 2016), or southern ITCZ index (e.g., Tian, 2015) cannot explain aspects of the double-ITCZ bias in the annual mean. Therefore, we construct a new, simpler double-ITCZ bias index to investigate the severity of the double-ITCZ pattern over the central Pacific in annual mean precipitation. We define a double-ITCZ bias index using the precipitation bias over the following three boxes: northern Pacific box (NB; 5°N–15°N, 160°E–120°W), Southern Pacific box (SB; 15°S–5°S, 160°E–120°W) and EB, such that

Double ITCZ bias = 
$$\frac{(NB + SB)}{2} - EB$$
 (1)

The magnitude of double-ITCZ bias index varies depending on the zonal boundary of these boxes, but the relationships presented in the paper do not (see Figure S1 for the impact of box size in zonal direction). See Figure 1 for illustrations of these boxes relative to time-mean SST and precipitation fields as well as biases and their dependence on one another. In this paper, we show the annual mean climatology of all fields throughout.

#### 2.2. AGCM Experiments

Atmospheric general circulation model (AGCM) experiments with prescribed SSTs were performed to examine the impact of different equatorial Pacific SST biases on tropical Pacific precipitation, thereby testing the hypothesis that SST biases regulate the magnitude of the simulated double ITCZ. The AGCM used in this study is ECHAM4.6 from the Max Planck Institute in Hamburg (HAM), a branch of the European Centre (EC) for Medium Range Weather Forecasts (Roeckner et al., 1996). We ran this model at T42 resolution ( $\sim$ 2.8° × 2.8°) with 19 vertical levels and daily outputs are obtained. Vertical diffusion is calculated based on turbulent kinetic energy using a high-order closure scheme and surface turbulent fluxes are calculated from Monin-Obukhov similarity theory. Cumulus parameterization of this model is based on the bulk mass flux concept (Tiedtke, 1989). ECHAM4.6 has been widely used to study atmospheric responses to tropical SST anomalies (Fu et al., 2003; Xiang et al., 2013; Zhang & Li, 2017; Zhang & Karnauskas, 2017).

We have performed seven experiments differing only in the prescribed monthly climatological SST fields. The control run was forced with the observed SST climatology (i.e., bias = 0 °C), and six experiments with idealized biases (BIAS) of -2.0, -1.5, -1.0, -0.5, 0.5, and 1.0 °C imposed over the domain 5°S–5°N, 150°E-82°W were also performed. While the magnitude of the SST bias imposed over the equatorial Pacific region varies in the six different BIAS experiments, the baseline SST climatology over other regions remains the same. To reduce the possible impact of spurious SST gradients along the boundaries of the prescribed biases, we applied a simple taper barrier with one half the magnitude of the SST anomaly instead of zero SST anomaly. For each AGCM experiment, the model was integrated for 30 years, and the first 5 years were discarded given that it takes a few years for this model (like most AGCMs) to reach an equilibrium state with the SST forcing (Zhang & Karnauskas, 2017). As with the CGCMs and observations, we only show the annual mean climatology in this study.



#### 2.3. Linear Regression Model

Finally, we used a simple linear regression model to explore the possible improvement of the predictability of precipitation trends based on SST biases and SST trends in CMIP5 models as predictors. Response  $(y_i)$  of *i*th CMIP5 model in a linear regression model can be expressed as below:

$$y_i = \beta_0 + \beta_1 X_{1i} + \beta_2 X_{2i} + \varepsilon_i \tag{2}$$

where  $\beta_0$  is the constant terms in the model,  $\beta_k(k = 1, 2)$  is the *k*th coefficient,  $X_{ki}$  (k = 1, 2) is the *i*th CMIP5 model on the *k*th predictor variable, and  $\varepsilon_i$  is the *i*th noise term, that is, random error. Here  $X_1$  and  $X_2$  are the SST bias and the SST trend over EB, respectively. Based on this regression model and using the SST bias and trend over the EB as the predictors, we calculate the estimated coefficients (i.e., linear model fit). Then, a predicted precipitation trend value over the EB is calculated using the linear model fit value and the predictor (SST bias and SST trend), based on trained discriminant analysis classification model fit. The predicted response is computed for the entire time span covered by the predictors, and it is assumed that the various classes generate data based on different Gaussian distributions (see Eisenbeis & Avery, 1972 for details about discriminant analysis classification).

## 3. Results

Regional details of tropical SST distributions (Figure 1b) and continental configuration have an important influence on the position and shape of the ITCZ. The simulated precipitation in CGCMs is not only biased, but the bias magnitude also varies from model to model (see Figures S2 and S3 for individual model bias in precipitation). The multimodel mean bias (Figures 1c and 1d) clearly shows the coexistence of double-ITCZ pattern along with equatorial Pacific cold tongue SST bias (see Figures S4 and S5 for individual model bias in SST). Tropical precipitation biases in CGCMs interact with SST biases generated by interaction with the ocean (Xiang et al., 2017). Regression of precipitation bias fields on SST bias averaged over the EB (Figure 1e) suggests that precipitation biases covary with SST bias over the EB and inversely vary with the off-equatorial zones where a double ITCZ exits. This result highlights the strong association between spatially coherent precipitation biases and equatorial SST biases, particularly over the western Pacific. However, a negative regression coefficient value over a large portion of the SB in Figure 1e may be related to the nonparametric nature of this regression calculation and involves intermodel differences. Since SST and atmospheric energy budget are strongly correlated near the equator, the link between the cold SST biases and excessive double ITCZ can be explained using the energy flux framework, which predicts stronger double ITCZ in the western Pacific (Adam et al., 2018).

A stronger cold SST bias appears to exacerbate the double-ITCZ bias (Figure 2a). Furthermore, precipitation trends in CGCMs are statistically related to the SST bias by a positive linear correlation (Figure 2b), implying (but not yet proving) a possible control of SST bias on precipitation trends. As the double-ITCZ bias includes a dry precipitation bias over the equatorial Pacific, SST biases are also closely related to an increase in dry precipitation bias over EB (Figure 2c). While the SST bias becomes positive (i.e., less severely biased), the equatorial dry precipitation bias is also alleviated. In agreement with these relationships, a severe double-ITCZ bias is associated with a lower precipitation trend over the EB. These results within mean-state biases and trends in CGCMs suggest possible interdependencies of precipitation trends with other biases as previously shown. We therefore hypothesize that the SST bias over the equatorial Pacific in CGCMs has an important role on the emergence of a double-ITCZ bias with implications for the simulated equatorial precipitation trend due to a strong relationship between SST and convection merely following the Clausius-Clapeyron equation. However, a severely cold SST bias diminishes the ability of the same incremental SST trend to influence the precipitation trend. In other words, for warmer initial SST, there is greater potential for rainfall to respond to further incremental changes in SST.

It is difficult to disentangle SST-precipitation relationships from coupled simulations due to the challenge of isolating the SST-forced atmosphere response and intrinsic variability of the atmosphere (He et al., 2018) and due to ocean-atmosphere coupling. In other words, correlation is not equivalent to causation. The previous discussion (Figure 2) provides statistical support to our hypothesis of the control of SST bias



**Figure 2.** Relationship among mean-state annual biases and trends in historical simulations. Scatterplots for (a) SST bias versus double-ITCZ bias, (b) SST bias versus precipitation trend (in  $mm \cdot day^{-1} \cdot century^{-1}$ ), (c) SST bias versus precipitation bias, and (d) double-ITCZ bias versus precipitation trend ( $mm \cdot day^{-1} \cdot century^{-1}$ ). Red and black dots denote each CMIP5 model and each AGCM experiments, respectively. SST biases and precipitation bias/double-ITCZ bias are in °C and  $mm \cdot day^{-1}$ , respectively.

over double-ITCZ bias and precipitation bias. To test the hypothesis more systematically, we carried out a set of idealized AGCM experiments as described in section 2. The AGCM experiments strongly agree (black dots in Figures 2a and 2c) with the CGCM's implied dependence of the double-ITCZ bias and the precipitation bias on the SST bias. However, the double-ITCZ bias becomes less sensitive in the BIAS experiments beyond -1 °C. Because of structural diversity across models, the mean state of each CGCM is different (red dots), whereas the AGCM experiments without ocean-atmosphere coupling and no changes in model configuration involve less complexity than CGCMs. Additionally, nonlinear double-ITCZ characteristics in AGCM experiments may be a consequence of reduced sensitivity to decreased (increased) precipitation bias over the EB (NB and SB) beyond -1 °C SST bias (black dots in Figure 2a). This nonlinear response of precipitation bias to SST bias needs further study. It is noteworthy that despite intermodel spread in the location of the southern lobe of the ITCZ (akin to the South Pacific Convergence Zone or SPCZ in the real world) in CMIP5 simulations, the multimodel mean bias pattern places the mean location toward the eastern Pacific. This may be related to the incomplete representation of ocean dynamics that have an important role in ocean-atmosphere coupling and trends in CGCM simulations (Coats & Karnauskas, 2018; Karnauskas et al., 2007; Karnauskas et al., 2012).

While the above analysis demonstrates the critical role of SST biases on precipitation biases over the tropical Pacific including those manifested as a double ITCZ, the spatial patterns of precipitation biases (Figure 3) show the response to different SST bias levels over the cold tongue region, and indicate the increase of severity of double-ITCZ pattern. Figure 3 indeed clearly demonstrates the crucial role of the



**Figure 3.** Annual mean precipitation bias (in  $mm \cdot day^{-1}$ ) from each AGCM experiments. Legends in each subplot denote the applied SST bias in SST forcing over equatorial Pacific. Bias level 0.0 indicates control experiment forced with observed SST.





**Figure 4.** Precipitation prediction using a linear regression model. Scatterplot between predicted centennial precipitation trend (in mm) and actual simulated centennial precipitation trend (in mm) from historical simulations. Predicted precipitation trend is calculated using SST bias and SST trend as the predictor. Black dots denote each CMIP5 model. The regression line is overlaid on the scatterplot. The correlation coefficient between predicted precipitation and actual precipitation is 0.43 at more than 95% confidence level.

SST bias over equatorial Pacific on the emergence of a double-ITCZ bias. We cannot rule out the possibility of the control of other potential factors on the emergence of the double ITCZ. Nonetheless, our results strongly suggest that SST bias has at least one of the crucial controls on the double-ITCZ bias. Consistent with several earlier studies (e.g., Hirota et al., 2011), the control experiment also suggests a possible emergence of a double-ITCZ pattern even without a SST bias. This implies that at least a substantial source of double-ITCZ pattern lies in the atmospheric component of CGCMs. Despite this fact, our results highlight how a mean-state equatorial SST bias can control the severity of the double-ITCZ pattern, indicating possible consequence of precipitation trends in CGCMs.

We further explore the sensitivity of precipitation biases to SST biases in CGCMs over the equatorial Pacific (Figure S6) and how it depends on the regime (i.e., warm pool versus cold tongue). We used longitude as a variable spanning such regimes, and calculated the bias dependency for several 10° longitude-wide zones across equatorial Pacific. Results clearly suggest that the sensitivity of precipitation bias to SST bias is greater over the western and central equatorial Pacific than eastern equatorial Pacific. This feature is also seen in Figure 1e. The higher sensitivity of precipitation over these regions of warmer mean SST is likely related to a greater change in surface pressure for the same incremental change in SST, again simply invoking on the Clausius-Clayperon equation. In general, the wet bias decreases (or dry bias increases) with an increase in cold bias (Figure S6). However, unlike a general characteristics precipitation bias increases (i.e., increases wet bias) up to the threshold SST bias of 0 °C

and then decreases with an decrease in the SST bias (i.e., increase in the cold SST bias) over the western Pacific warm pool region (140°E–150°E box), which may be related to additional responses of landatmosphere interactions and weak SST-convection relationship over this region. Contrary to the central and eastern Pacific, the SST-convection relationship over the western Pacific region is largely controlled by the large-scale dynamics, rather than by local thermodynamics (Waliser & Graham, 1993; Wu & Moncrieff, 1999). Earlier studies reported that deep convection over western Pacific warm pool region (where maximum SSTs occurs) is not correlated with SST variations over the same region (Fu et al., 1992), even inversely related with the increase of SST beyond a certain threshold (Waliser & Graham, 1993).

The above results directly imply that improved prediction of tropical precipitation trends may be possible with simple knowledge of model-dependent tropical climate biases. Using a linear regression model and considering spatially averaged SST biases and trends over the EB region as the only predictors, we calculate a predicted value of precipitation trends that are significantly correlated (correlation coefficient 0.43 at more than 95% confidence level) with the actual simulated precipitation trend (Figure 4). This is not the case if only the SST trend is used as the predictor (correlation coefficient 0.25) or only the precipitation bias is used as the predictor (correlation coefficient 0.31). It is noteworthy that this approach is not likely to explain the entire range of interactions between mean-state biases and precipitation trends, as the CMIP5 models are fully coupled and have important structural differences. Nonetheless, this result suggests a possible way to adjust or constrain the simulated precipitation trends in an "off-line" way according to their dependence on mean-state biases described herein.

### 4. Summary and Conclusion

Using historical simulations from 24 CMIP5 climate models, we explored the potential relationship between the double-ITCZ and equatorial SST biases in the tropical Pacific sector, and its implications for tropical precipitation trends in CGCMs. Our results suggest that a cold equatorial SST bias linked to the emergence of a double-ITCZ bias in CGCMs. Due to a strong SST-convection relationship over the tropical Pacific, the double-ITCZ bias in CGCMs is sensitive to background SST, which is further supported by a set of AGCM experiments targeted at this hypothesis. The AGCM experiments with different SST biases demonstrate this clear dependence.

SST and precipitation over the equatorial Pacific region are critical for a multitude of factors, not only for CGCM biases but also for its impact on climate over various regions of the globe through remote teleconnections. For example, equatorial Pacific SST is strongly connected with El Niño–Southern Oscillation diversity, which is further known to have a profound influence on the weather and climate around the globe (e.g., McPhaden et al., 2006). Moreover, several recent studies reported the increase in the frequency of central Pacific El Niño events in a climate-changing scenario (e.g., Lee & McPhaden, 2010). Furthermore, as global warming-induced precipitation changes are likely to occur mostly over equatorial Pacific Ocean (Zhou & Xie, 2015), it is meaningful to have a better understanding of relationships between the SST and precipitation biases in CGCM simulations and to improve the simulated precipitation trends in CGCMs.

The presented results highlight not only the important implications of SST biases on historical simulation of precipitation but also the likely impact on future precipitation projections. Our study provides a simple framework for reducing uncertainty in rainfall projections. It may be possible to increase the reliability of precipitation trends in CMIP5 models given such mean-state bias information, all of which is available prior to CGCM biases being completely resolved by the model development community.

#### References

- Adam, O., Bischoff, T., & Schneider, T. (2016). Seasonal and interannual variations of the energy flux equator and ITCZ. Part II: Zonally varying shifts of the ITCZ. Journal of Climate, 29, 7281–7293. https://doi.org/10.1175/JCLI-D-15-0710.1
- Adam, O., Schneider, T., & Brient, F. (2018). Regional and seasonal variations of the double-ITCZ bias in CMIP5 models. *Climate Dynamics*, *51*, 101–117. https://doi.org/10.1007/s00382-017-3909-1
- Adam, O., Schneider, T., Brient, F., & Bischoff, T. (2016). Relation of the double-ITCZ bias to the atmospheric energy budget in climate models. *Geophysical Research Letters*, 43, 7670–7677. https://doi.org/10.1002/2016GL069465
- Adler, R. F., Huffman, G. J., Chang, A., Ferraro, R., Xie, P. P., Janowiak, J., et al. (2003). The version-2 Global Precipitation Climatology Project (GPCP) monthly precipitation analysis (1979–present). *Journal of Hydrometeorology*, 4(6), 1147–1167. https://doi.org/10.1175/ 1525-7541(2003)004<1147:TVGPCP>2.0.CO;2
- Arora, V. K., Scinocca, J. F., Boer, G. J., Christian, J. R., Denman, K. L., Flato, G. M., et al. (2011). Carbon emission limits required to satisfy future representative concentration pathways of greenhouse gases. *Geophysical Research Letters*, 38, L05805. https://doi.org/10.1029/ 2010GL046270
- Bellucci, A., Gualdi, S., & Navarra, A. (2010). The double-ITCZ syndrome in coupled general circulation models: The role of large-scale vertical circulation regimes. *Journal of Climate*, 23, 1127–1145. https://doi.org/10.1175/2009JCLI3002.1
- Bentsen, M., Bethke, I., Debernard, J. B., Iversen, T., Kirkevåg, A., Seland, Ø., et al. (2013). The Norwegian earth system model, NorESM1-M—Part 1: Description and basic evaluation of the physical climate. *Geoscientific Model Development*, 6, 687–720. https://doi.org/ 10.5194/gmd-6-687-2013
- Bi, D., Dix, M., Marsland, S. J., O'Farrell, S., Rashid, H., Uotila, P., et al. (2013). The ACCESS coupled model: Description, control climate and evaluation. Australian Meteorological and Oceanographic Journal, 63, 41–64. https://doi.org/10.22499/2.6301.004
- Chadwick, R., Boutle, I., & Martin, G. (2013). Spatial patterns of precipitation change in CMIP5: Why the rich do not get richer in the tropics. *Journal of Climate*, *26*, 3803–3822. https://doi.org/10.1175/JCLI-D-12-00543.1
- Coats, S., & Karnauskas, K. B. (2018). A role for the equatorial undercurrent in the ocean dynamical thermostat. *Journal of Climate*, *31*, 6245–6261. https://doi.org/10.1175/JCLI-D-17-0513.1
- De Szoeke, S. P., & Xie, S. P. (2008). The tropical eastern Pacific seasonal cycle: Assessment of errors and mechanisms in IPCC AR4 coupled ocean-atmosphere general circulation models. *Journal of Climate*, *21*, 2573–2590. https://doi.org/10.1175/2007JCLI1975.1
- Donner, L. J., Wyman, B. L., Hemler, R. S., Horowitz, L. W., Ming, Y., Zhao, M., et al. (2011). The dynamical core, physical parameterizations, and basic simulation characteristics of the atmospheric component AM3 of the GFDL global coupled model CM3. *Journal of Climate*, 24(13), 3484–3519. https://doi.org/10.1175/2011JCLI3955.1
- Dufresne, J. L., Foujols, M. A., Denvil, S., Caubel, A., Marti, O., Aumont, O., et al. (2013). Climate change projections using the IPSL-CM5 Earth system model: From CMIP3 to CMIP5. *Climate Dynamics*, 40, 2123–2165. https://doi.org/10.1007/s00382-012-1636-1
- Eisenbeis, R. A., & Avery, R. B. (1972). Discriminant Analysis and Classification Procedures: Theory and Applications. Lexington, MA: DC Heath.
- Fu, R., Del Genio, A. D., Rossow, W. B., & Liu, W. T. (1992). Cirrus-cloud thermostat for tropical sea surface temperatures tested using satellite data. Nature, 358(6385), 394. https://doi.org/10.1038/358394a0
- Fu, X., Wang, B., Li, T., & McCreary, J. P. (2003). Coupling between northward-propagating, intraseasonal oscillations and sea surface temperature in the Indian Ocean. *Journal of the Atmospheric Sciences*, 60(15), 1733–1753. https://doi.org/10.1175/1520-0469(2003)060<1733:CBNIOA>2.0.CO;2
- Gent, P. R., Danabasoglu, G., Donner, L. J., Holland, M. M., Hunke, E. C., Jayne, S. R., et al. (2011). The Community Climate System Model version 4. Journal of Climate, 24, 4973–4991. https://doi.org/10.1175/2011JCL14083.1
- Gill, A. (1980). Some simple solutions for heat-induced tropical circulation. Quarterly Journal of the Royal Meteorological Society, 106(449), 447–462. https://doi.org/10.1002/qj.49710644905
- Gu, G., Adler, R. F., & Sobel, A. H. (2005). The eastern Pacific ITCZ during the boreal spring. Journal of the Atmospheric Sciences, 62(4), 1157–1174. https://doi.org/10.1175/JAS3402.1
- Guilyardi, E., Delecluse, P., Gualdi, S., & Navarra, A. (2003). Mechanisms for ENSO phase change in a coupled GCM. Journal of Climate, 16(8), 1141–1158. https://doi.org/10.1175/1520-0442(2003)16<1141:MFEPCI>2.0.CO;2

#### Acknowledgments

This work was funded by Singapore Ministry of Education Academic Research Fund Tier 2 (project MOE2016-T2-1-016). We acknowledge the WCRP Working Group on Coupled Modeling and U.S. DOE/PCMDI for CMIP, and thank the climate modeling groups (listed in Table S1 of this paper) for producing and making available their model output (http://cmip-pcmdi. llnl.gov/cmip5/). GPCP Precipitation data were provided by the NOAA/OAR/ESRL PSD, Boulder, CO, USA, from their Web site at https:// www.esrl.noaa.gov/psd/. Climatological precipitation data from AGCM simulations are available at https://figshare.com/s/

5cb953e846dbceaeaa60.



- Hawcroft, M., Haywood, J. M., Collins, M., Jones, A., Jones, A. C., & Stephens, G. (2017). Southern Ocean albedo, inter-hemispheric energy transports and the double ITCZ: Global impacts of biases in a coupled model. *Climate Dynamics*, 48, 2279–2295. https://doi.org/10.1007/ s00382-016-3205-5
- He, J., Johnson, N. C., Vecchi, G. A., Kirtman, B., Wittenberg, A. T., & Sturm, S. (2018). Precipitation sensitivity to local variations in tropical sea surface temperature. *Journal of Climate*, 31, 9225–9238. https://doi.org/10.1175/JCLI-D-18-0262.1
- Held, I. M., & Soden, B. J. (2006). Robust responses of the hydrological cycle to global warming. Journal of Climate, 19, 5686–5699. https://doi.org/10.1175/JCLI3990.1
- Hirota, N., Takayabu, Y. N., Watanabe, M., & Kimoto, M. (2011). Precipitation reproducibility over tropical oceans and its relationship to the double ITCZ problem in CMIP3 and MIROC5 climate models. *Journal of Climate*, 24, 4859–4873. https://doi.org/10.1175/ 2011JCLI4156.1
- Hwang, Y. T., & Frierson, D. M. (2013). Link between the double-Intertropical Convergence Zone problem and cloud biases over the Southern Ocean. *Proceedings of the National Academy of Sciences*, *110*, 4935–4940. https://doi.org/10.1073/pnas.1213302110
- Inness, P. M., & Slingo, J. M. (2003). Simulation of the Madden–Julian Oscillation in a coupled general circulation model. Part I: Comparison with observations and an atmosphere-only GCM. *Journal of Climate*, 16(3), 345–364. https://doi.org/10.1175/1520-0442(2003)016<0345:SOTMJO>2.0.CO;2
- Karnauskas, K. B., Johnson, G. C., & Murtugudde, R. (2012). An equatorial ocean bottleneck in global climate models. Journal of Climate, 25, 343–349. https://doi.org/10.1175/JCLI-D-11-00059.1
- Karnauskas, K. B., Murtugudde, R., & Busalacchi, A. J. (2007). The effect of the Galápagos Islands on the equatorial Pacific cold tongue. Journal of Physical Oceanography, 37, 1266–1281. https://doi.org/10.1175/JPO3048.1
- Kay, J. E., Wall, C., Yettella, V., Medeiros, B., Hannay, C., Caldwell, P., & Bitz, C. (2016). Global climate impacts of fixing the Southern Ocean shortwave radiation bias in the Community Earth System Model (CESM). Journal of Climate, 29, 4617–4636. https://doi.org/ 10.1175/JCLI-D-15-0358.1
- Kent, C., Chadwick, R., & Rowell, D. P. (2015). Understanding uncertainties in future projections of seasonal tropical precipitation. Journal of Climate, 28(11), 4390–4413.
- Lee, T., & McPhaden, M. J. (2010). Increasing intensity of El Niño in the central-equatorial Pacific. *Geophysical Research Letters*, 37, L14603. https://doi.org/10.1029/2010GL044007
- Li, G., & Xie, S. P. (2014). Tropical biases in CMIP5 multimodel ensemble: The excessive equatorial Pacific cold tongue and double ITCZ problems. Journal of Climate, 27, 1765–1780. https://doi.org/10.1175/JCLI-D-13-00337.1
- Lietzke, C. E., Deser, C., & Vonder Haar, T. H. (2001). Evolutionary structure of the eastern Pacific double ITCZ based on satellite moisture profile retrievals. Journal of Climate, 14(5), 743–751. https://doi.org/10.1175/1520-0442(2001)014<0743:ESOTEP>2.0.CO;2

Lin, J. L. (2007). The double-ITCZ problem in IPCC AR4 coupled GCMs: Ocean-atmosphere feedback analysis. Journal of Climate, 20, 4497-4525. https://doi.org/10.1175/JCLI4272.1

- Long, M. C., Lindsay, K., Peacock, S., Moore, J. K., & Doney, S. C. (2013). Twentieth-century oceanic carbon uptake and storage in CESM1 (BGC). Journal of Climate, 26, 6775–6800. https://doi.org/10.1175/JCLI-D-12-00184.1
- Ma, C. C., Mechoso, C. R., Robertson, A. W., & Arakawa, A. (1996). Peruvian stratus clouds and the tropical Pacific circulation: A coupled ocean-atmosphere GCM study. *Journal of Climate*, 9(7), 1635–1645. https://doi.org/10.1175/1520-0442(1996)009<1635:PSCATT>2.0. CO;2
- Marshall, J., Donohoe, A., Ferreira, D., & McGee, D. (2014). The ocean's role in setting the mean position of the Inter-Tropical Convergence Zone. *Climate Dynamics*, 42, 1967–1979. https://doi.org/10.1007/s00382-013-1767-z
- Martin, G. M., Ringer, M. A., Pope, V. D., Jones, A., Dearden, C., & Hinton, T. J. (2006). The physical properties of the atmosphere in the new Hadley Centre global environmental model (HadGEM1). Part I: Model description and global climatology. *Journal of Climate*, 19, 1274–1301. https://doi.org/10.1175/JCLI3636.1

McPhaden, M. J., Zebiak, S. E., & Glantz, M. H. (2006). ENSO as an integrating concept in earth science. *Science*, *31*(5806), 1740–1745. Möbis, B., & Stevens, B. (2012). Factors controlling the position of the Intertropical Convergence Zone on an aquaplanet. *Journal of* 

- Advances in Modeling Earth Systems, 4, M00A04. https://doi.org/10.1029/2012MS000199
- Neelin, J. D., & Held, I. M. (1987). Modeling tropical convergence based on the moist static energy budget. *Monthly Weather Review*, 115(1), 3–12. https://doi.org/10.1175/1520-0493(1987)115<0003:MTCBOT>2.0.CO;2
- Oueslati, B., & Bellon, G. (2015). The double ITCZ bias in CMIP5 models: Interaction between SST, large-scale circulation and precipitation. Climate Dynamics, 44, 585–607. https://doi.org/10.1007/s00382-015-2468-6
- Popp, M., & Lutsko, N. J. (2017). Quantifying the zonal-mean structure of tropical precipitation. *Geophysical Research Letters*, 44, 9470–9478. https://doi.org/10.1002/2017GL075235
- Power, S., Delage, F., Chung, C., Kociuba, G., & Keay, K. (2013). Robust twenty-first-century projections of El Niño and related precipitation variability. *Nature*, 502, 541–545. https://doi.org/10.1038/nature12580
- Qin, Y., & Lin, Y. (2018). Alleviated double ITCZ problem in the NCAR CESM1: A new cloud scheme and the working mechanisms. Journal of Advances in Modeling Earth Systems, 10, 2318–2332. https://doi.org/10.1029/2018MS001343
- Raddatz, T. J., Reick, C. H., Knorr, W., Kattge, J., Roeckner, E., Schnur, R., et al. (2007). Will the tropical land biosphere dominate the climate–carbon cycle feedback during the twenty-first century? *Climate Dynamics*, 29, 565–574. https://doi.org/10.1007/s00382-007-0247-8
- Richter, I., Doi, T., Chang, P., Xu, Z., Kataoka, T., Tozuka, T., et al. (2016). An overview of coupled GCM biases in the tropics. In *Indo-Pacific Climate Variability and Predictability* (pp. 213–263). Texas: World Scientific. https://doi.org/10.1142/9789814696623\_0008
- Roeckner, E., Arpe, K., Begntsson, L., Christoph, M., Claussen, M., Dumenil, L., et al. (1996). The atmospheric general circulation model ECHAM-4: Model description and simulation of present-day climate, Max Planck Institute for Meteorology Rep., 218(90).
- Rotstayn, L. D., Jeffrey, S. J., Collier, M. A., Dravitzki, S. M., Hirst, A. C., Syktus, J. I., & Wong, K. K. (2012). Aerosol-induced changes in summer rainfall and circulation in the Australasian region: A study using single-forcing climate simulations. *Atmospheric Chemistry and Physics Discussions*, 12, 5107–5188. https://doi.org/10.5194/acpd-12-5107-2012
- Schneider, T., Bischoff, T., & Haug, G. H. (2014). Migrations and dynamics of the intertropical convergence zone. Nature, 513(7516), 45.
- Scoccimarro, E., Gualdi, S., Bellucci, A., Sanna, A., Giuseppe Fogli, P., Manzini, E., et al. (2011). Effects of tropical cyclones on ocean heat transport in a high-resolution coupled general circulation model. *Journal of Climate*, 24, 4368–4384. https://doi.org/10.1175/ 2011JCLI4104.1
- Seager, R., Naik, N., & Vecchi, G. A. (2010). Thermodynamic and dynamic mechanisms for large-scale changes in the hydrological cycle in response to global warming. *Journal of Climate*, 23, 4651–4668. https://doi.org/10.1175/2010JCLI3655.1

Siongco, A., Hohenegger, C., & Stevens, B. (2015). The Atlantic ITCZ bias in CMIP5 models. Climate Dynamics, 45(5-6), 1169-1180.



Song, F., & Zhang, G. J. (2016). Effects of southeastern Pacific Sea surface temperature on the double-ITCZ bias in NCAR CESM1. Journal of Climate, 29, 7417–7433. https://doi.org/10.1175/JCLI-D-15-0852.1

Song, X., & Zhang, G. J. (2018). The roles of convection parameterization in the formation of double ITCZ syndrome in the NCAR CESM: I. Atmospheric processes. *Journal of Advances in Modeling Earth Systems*, *10*, 842–866. https://doi.org/10.1002/2017MS001191

Taylor, K. E., Stouffer, R. J., & Meehl, G. A. (2012). An overview of CMIP5 and the experiment design. Bulletin of the American Meteorological Society, 93, 485–498. https://doi.org/10.1175/BAMS-D-11-00094.1

- Tian, B. (2015). Spread of model climate sensitivity linked to double-Intertropical Convergence Zone bias. *Geophysical Research Letters*, 42, 4133–4141. https://doi.org/10.1002/2015GL064119
- Tiedtke, M. I. C. H. A. E. L. (1989). A comprehensive mass flux scheme for cumulus parameterization in large-scale models. *Monthly Weather Review*, 117(8), 1779–1800. https://doi.org/10.1175/1520-0493(1989)117<1779:ACMFSF>2.0.CO;2
- Trenberth, K. E., Branstator, G. W., Karoly, D., Kumar, A., Lau, N. C., & Ropelewski, C. (1998). Progress during TOGA in understanding and modeling global teleconnections associated with tropical sea surface temperatures. *Journal of Geophysical Research*, 103(C7), 14,291–14,324. https://doi.org/10.1029/97JC01444

Vecchi, G. A., & Soden, B. J. (2007). Global warming and the weakening of the tropical circulation. Journal of Climate, 20, 4316–4340. https://doi.org/10.1175/JCLI4258.1

Voldoire, A., Sanchez-Gomez, E. D. Salas y Mélia, Salas y Mélia, D, Decharme, B., Cassou, C., Sénési, S., et al. (2013). The CNRM-CM5. 1 global climate model: Description and basic evaluation. *Climate Dynamics*, 40(9–10), 2091–2121. https://doi.org/10.1007/s00382-011-1259-y

Volodin, E. M., Dianskii, N. A., & Gusev, A. V. (2010). Simulating present-day climate with the INMCM4. 0 coupled model of the atmospheric and oceanic general circulations. *Izvestiya, Atmospheric and Oceanic Physics*, 46, 414–431. https://doi.org/10.1134/ S000143381004002X

Waliser, D. E., & Gautier, C. (1993). A satellite-derived climatology of the ITCZ. Journal of Climate, 6(11), 2162–2174. https://doi.org/ 10.1175/1520-0442(1993)006<2162:ASDCOT>2.0.CO;2

Waliser, D. E., & Graham, N. E. (1993). Convective cloud systems and warm-pool sea surface temperatures: Coupled interactions and selfregulation. Journal of Geophysical Research, 98(D7), 12,881–12,893. https://doi.org/10.1029/93JD00872

Wang, C. C., Lee, W. L., Chen, Y. L., & Hsu, H. H. (2015). Processes leading to double Intertropical Convergence Zone bias in CESM1/CAM5. *Journal of Climate*, *28*, 2900–2915. https://doi.org/10.1175/JCLI-D-14-00622.1

Watanabe, S., Hajima, T., Sudo, K., Nagashima, T., Takemura, T., Okajima, H., et al. (2011). MIROC-ESM 2010: Model description and basic results of CMIP5-20c3m experiments. Geoscientific Model Development, 4, 845–872. https://doi.org/10.5194/gmd-4-845-2011

Wittenberg, A. T., Rosati, A., Lau, N. C., & Ploshay, J. J. (2006). GFDL's CM2 global coupled climate models. Part III: Tropical Pacific climate and ENSO. Journal of Climate, 19, 698–722. https://doi.org/10.1175/JCLI3631.1

Wu, X., & Moncrieff, M. W. (1999). Effects of sea surface temperature and large-scale dynamics on the thermodynamic equilibrium state and convection over the tropical western Pacific. *Journal of Geophysical Research*, 104(D6), 6093–6100. https://doi.org/10.1029/ 1998JD200116

Xiang, B., Wang, B., & Li, T. (2013). A new paradigm for the predominance of standing central Pacific warming after the late 1990s. Climate Dynamics, 41, 327–340. https://doi.org/10.1007/s00382-012-1427-8

- Xiang, B., Zhao, M., Held, I. M., & Golaz, J. C. (2017). Predicting the severity of spurious "double ITCZ" problem in CMIP5 coupled models from AMIP simulations. *Geophysical Research Letters*, 44, 1520–1527. https://doi.org/10.1002/2016GL071992
- Xiang, B., Zhao, M., Ming, Y., Yu, W., & Kang, S. M. (2018). Contrasting impacts of radiative forcing in the Southern Ocean versus southern tropics on ITCZ position and energy transport in one GFDL climate model. *Journal of Climate*, 31, 5609–5628. https://doi.org/10.1175/ JCLI-D-17-0566.1

Xie, S. P., Deser, C., Vecchi, G. A., Ma, J., Teng, H., & Wittenberg, A. T. (2010). Global warming pattern formation: Sea surface temperature and rainfall. Journal of Climate, 23, 966–986. https://doi.org/10.1175/2009JCLI3329.1

Xin, X.-G., Wu, T.-W., Li, J.-L., Wang, Z.-Z., Li, W.-P., Wu, F.-H., & U, W. (2013). How well does BCC\_CSM1. 1 reproduce the 20th century climate change over China? Atmospheric and Oceanic Science Letters, 6, 21–26. https://doi.org/10.1080/16742834.2013.11447053

Yang, M., Zhang, G. J., & Sun, D. Z. (2018). Precipitation and moisture in four leading CMIP5 models: Biases across large-scale circulation regimes and their attribution to dynamic and thermodynamic factors. *Journal of Climate*, 31, 5089–5106. https://doi.org/10.1175/JCLI-D-17-0718.1

Yongqiang, Y., Xuehong, Z., & Yufu, G. (2004). Global coupled ocean-atmosphere general circulation models in LASG/IAP. Advances in Atmospheric Sciences, 21, 444–455. https://doi.org/10.1007/BF02915571

Yukimoto, S., Adachi, Y., Hosaka, M., Sakami, T., Yoshimura, H., Hirabara, M., et al. (2012). A new global climate model of the meteorological research institute: MRI-CGCM3—Model description and basic performance. *Journal of the Meteorological Society of Japan. Ser. II*, 90, 23–64. https://doi.org/10.2151/jmsj.2012-A02

Zhang, C. (2001). Double ITCZs. Journal of Geophysical Research, 106(D11), 11,785–11,792. https://doi.org/10.1029/2001JD900046

Zhang, L., & Karnauskas, K. B. (2017). The role of tropical interbasin SST gradients in forcing Walker circulation trends. *Journal of Climate*, 30, 499–508. https://doi.org/10.1175/JCLI-D-16-0349.1

Zhang, L., & Li, T. (2017). Relative roles of differential SST warming, uniform SST warming and land surface warming in determining the Walker circulation changes under global warming. *Climate Dynamics*, 48, 987–997. https://doi.org/10.1007/s00382-016-3123-6

Zhang, X., Liu, H., & Zhang, M. (2015). Double ITCZ in coupled ocean-atmosphere models: From CMIP3 to CMIP5. *Geophysical Research Letters*, *42*, 8651–8659. https://doi.org/10.1002/2015GL065973

Zhou, Z. Q., & Xie, S. P. (2015). Effects of climatological model biases on the projection of tropical climate change. *Journal of Climate, 28*, 9909–9917. https://doi.org/10.1175/JCLI-D-15-0243.1